Author's response to reviews

Title: HIV Prevention for South African Youth: Systematic Review of the Evidence for 'What Works'

Authors:

Abigail Harrison (abigail_harrison@brown.edu)
Marie-Louise Newell (mnewell@africacentre.ac.za)
John Imrie (jimrie@africacentre.ac.za)
Graeme Hoddinott (ghoddinott@africacentre.ac.za)

Version: 2 Date: 3 October 2009

Author's response to reviews:

Revision in response to comments from Scientific Editor

Dear Ms Titmus,

In response to your email of 28 September, we have made the minor revisions requested to the title. This now reflects that the paper is a systematic review. In addition, we have ensured that our review meets the guidelines outlined in the PRISMA statement. Authors' full names and email addresses have also been added to the title page.

Thank you for your consideration of this manuscript.

Regards,

Abigail Harrison
Corresponding Author